


To: 'Kliphuis, Trais'; 'peake.tom@epa.gov'; 'Edwards, Jonathan'; 'Walsh, Jonathan'; 'Perrin, Alan’;
'Bob.Kehrman@wipp.ws'; 'Rick.Chavez@wipp.ws'; 'Stone.Nick@epa.gov'; ‘coleman.smith@state.nm.us’;
'brozowski.george@epa.goV'; 'Fraass, Ron'; 'Russell Hardy'; 'Veal.Lee@epamail.epa.gov'; 'Economy, Kathleen
(Economy.Kathleen@epa.gov)'; 'Poppell, Sam W. (Poppell.Sam@epa.gov)'

Cc: George Basabilvazo - WIPPNet; 'Reynolds, Tammy - NWP (Tammy.Reynolds@wipp.ws)'; 'Pace, Berry
(Berry.Pace@wipp.ws)'; ‘Alton.Harris@em.doe.gov'; Susan McCauslin; 'Joe Harvill (jharvill@portageinc.con ; 'Kennedy,
Scott - NWP (Scott.Kennedy@wipp.ws)'; 'Jones, Stewart - RES'; 'Berta QOates'; 'schultheisz.daniel@epa.gov; Philip Theisen
- ORISE; 'Dale Bignell'; Russ Patterson - WIPPNet; 'Kouba, Steve - WRES (Steve.Kouba@wipp.ws)'; Roget  2lson -
WIPPNet

Subject: WIPP Information - For Call Today

Attached is the action item listing, station A/B data, environmental samples tables, and the bioassay table.

Below is @a summary of the dis  ssi oin today: (Please note, activities and dates are subject to ¢ nge. Please
verify the most current dates ur any wuorm........ provided).

e (Cleaning the soot from the fire out of the waste hoist tower is beginning today with the floors.

e Water sample results from the rain event on 3/16 were received. All were less than MDC.

e Dr. Albert Wiley, Director REAC/TS is on-site today and tomorrow conducting an independent review of the
bioassay program and reviewing completed dose calculations. There are 12 urine samples and 3 fecal samples
currently in the lab. The last WBC is scheduled for 4/22. The only remaining dose calculation (pe ing any more
positive results) is related to the one positive urine sample. There are two samples yet to be submitted and 58
letters remaining to be issued. The target for completing all letters to the 149 personnel is 4/30.

e Station A and B Sampling: The CAM at Station B is functioning. The safety basis strategy for the CAM utilization is
being modified and will be resubmitted to DOE 4/9. Discussions are continuing regarding the CAM set
point. The system is expected to go into service on 4/18. NWP hopes to return approximately 90 additional
workers to the site on 4/21. (No change)

e SSand AlS Shaft/Rope Inspections: The weekly MSHA compliance inspection for the AIS did not occur yesterday
and is planned for today. The inspection of the SS will be done late today or tomorrow.

e Contamination survey to the waste face (Phase 3): This is being broken into multiple activities:

= Activity 1: Survey and proceed to airlock between W30 & E140 and to 1 10/S2000,
scheduled for 4/12 or early next week

= Activity 2: Survey and proceed to E140/52520 to evaluate future safe operating
base at the Waste Handling Lunchroom

= Activity 3: Establish safe operating base in WH lunchroom and survey to
W170/52750

Note: Activities 2 and 3 may be performed during the same entry, depending on contamination

conditions
= Activity 4: Identify source at Panel 7 Waste Face
Or Identify source at Panel 6
e The filter change mock-up is planned for the week of April 14™. NWP has completed a critical review the filter
replacement schedule and determined that they can be task ready on 5/12 (ten days earlier). The critical path,
which is to provide makeup air for the fan. 34 (2 failed testing) HEPA filters are expected to arrive on 4/9. An
additional 60 filters are expected on 4/17.
e Shipments of TRU waste from LANL to WCS are continuing. EPA is visiting WCS today.

A discussion will be held regarding the graphs that were sent out yesterday at the end of the meeting. The color code is
as follows:
e Cyan =1 Pre-Filter MOD



e Green =2 Pre-Filter HIGH

e Yellow = 1* HEPA Bank

e Red = 2" HEPA Bank
As a reminder, the call-in number is (866) 723-6758, code 8040901#
Thanks

Oba












